Authors's response to reviews

Title: Design and respondent selection of a population-based study on associations between breast cancer screening, lifestyle and quality of life

Authors:

Tytti Sarkeala (tytti.sarkeala@cancer.fi)
Sirpa Heinävaara (sirpa.heinavaara@cancer.fi)
Jonna Fredman (jonna.fredman@gmail.com)
Satu Männistö (satu.mannisto@thl.fi)
Riitta Luoto (riitta.luoto@uta.fi)
Maija Jäntti (maija.jantti@cancer.fi)
Nea Malila (nea.malila@cancer.fi)

Version: 4 Date: 4 December 2015

Author's response to reviews:

COVER LETTER

Dear Editor,

Please find attached our revised manuscript 'Design and respondent selection of a population-based study on associations between breast cancer screening, lifestyle and quality of life', and comments to the reviewer below this letter.

We have added information on permissions related to the study (p 8, lines 159-161), otherwise the manuscript is similar to the previous version.

On the behalf of the working group,

Tytti Sarkeala
The corresponding author
Finnish Cancer Registry
Unioninkatu 20–22
00130 Helsinki
Finland
tytti.sarkeala@cancer.fi
tel +358 50 411 4238